\begin{thebibliography}{10}

\bibitem{shaw84}
M.~C. Shaw,
\newblock {\em {Metal Cutting Principles}},
\newblock Oxford Series on Advanced Manufacturing, Oxford University Press,
  1984.

\bibitem{reuben98}
R.~Reuben,
\newblock {The Role of Acoustic Emission in Industrial Condition Monitoring},
\newblock International Journal of COMADEM {\bf 1}(4), 35--46 (October 1998).

\bibitem{inasaki81}
I.~Inasaki and S.~Yonetsu,
\newblock {In-Process Detection of Cutting Tool Damage by Acoustic Emission
  Measurement},
\newblock in {\em Proceedings of the 22nd International Machine Tool Design and
  Research Conference}, pages 261--268, Manchester, 1981.

\bibitem{lan82}
M.~S. Lan and D.~A. Dornfeld,
\newblock {Experimental Studies of Tool Wear via Acoustic Emission Analysis},
\newblock in {\em Proceedings of the 10th North American Manufacturing Research
  Conference}, pages 305--311, May 1982.

\bibitem{messaritis86}
V.~Messaritis and W.~K.~D. Borthwick,
\newblock {Processing Acoustic Emission Signal Data for Characterising Cutting
  Tool Wear and Chip Management},
\newblock in {\em Proceeding of the 1st International Conference on
  Computer-Aided Production Engineering}, pages 261--268, 1986.

\bibitem{naerheim88}
Y.~Naerheim and M.~S. Lan,
\newblock {Acoustic Emission Reveals Information about the Metal Cutting
  Process and Tool Wear},
\newblock in {\em Proceedings of the 16th North American Manufacturing Research
  Conference}, pages 240--244, May 1988.

\bibitem{blum90}
T.~Blum and I.~Inasaki,
\newblock {A Study on Acoustic Emission from the Orthogonal Cutting Process},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  112}(3), 203--211 (August 1990).

\bibitem{dimla00}
D.~E. Dimla, Snr.,
\newblock {Sensor Signals for Tool-Wear Monitoring in Metal Cutting
  Operations---A Review of Methods},
\newblock International Journal of Machine Tools and Manufacture {\bf 40},
  1073--1098 (2000).

\bibitem{li02}
X.~Li,
\newblock {A brief Review: Acoustic Emission Method for Tool Wear Monitoring
  during Turning},
\newblock International Journal of Machine Tools and Manufacture {\bf 42},
  157--165 (2002).

\bibitem{moriwaki80}
T.~Moriwaki,
\newblock {Detection of Cutting Tool Fracture by Acoustic Emission
  Measurement},
\newblock Annals of the CIRP {\bf 29}(1), 35--40 (1980).

\bibitem{lan84}
M.~S. Lan and D.~A. Dornfeld,
\newblock {In-Process Tool Fracture Detection},
\newblock Journal of Engineering Materials and Technology---Transactions of the
  ASME {\bf 106}(2), 111--118 (April 1984).

\bibitem{diei87}
E.~N. Diei and D.~A. Dornfeld,
\newblock {A Model of Tool Fracture Generated Acoustic Emission during
  Machining},
\newblock Journal of Engineering for Industry---Transactions of ASME {\bf
  109}(3), 227--233 (August 1987).

\bibitem{haili03}
W.~Haili, S.~Hua, C.~Ming, and H.~Dejin,
\newblock {On-Line Tool Breakage Monitoring in Turning},
\newblock Journal of materials Processing Technology {\bf 139}, 237--242
  (2003).

\bibitem{dornfeld83}
D.~A. Dornfeld and M.~S. Lan,
\newblock {Chip Form Detection using Acoustic Emission},
\newblock in {\em Proceedings of the 11th North American Manufacturing Research
  Conference}, pages 386--389, Madison, May 1983.

\bibitem{uehara84}
K.~Uehara and Y.~Kanda,
\newblock {Indentification of Chip Formation Mechanism through Acoustic
  Emission Measurements},
\newblock Annals of the CIRP {\bf 33}(1), 71--74 (1984).

\bibitem{dornfeld85}
D.~A. Dornfeld and C.~S. Pan,
\newblock {A Study of Continuous/Discontinuous Chip Formation using Acoustic
  Emission},
\newblock Journal of Applied Metal Working {\bf 4}(1), 18--29 (July 1985).

\bibitem{iturrospe05}
A.~Iturrospe, D.~A. Dornfeld, V.~Atxa, and J.~M. Abete,
\newblock {Bicepstrum Based Blind Identification of the Acoustic Emission (AE)
  Signal in Precisison Turning},
\newblock Mechanical Systems and Signal Processing {\bf 19}, 447--466 (2005).

\bibitem{oxley89}
P.~L.~B. Oxley,
\newblock {\em {Mechanics of Machining: An Analytical Approach to Assessing
  Machinability}},
\newblock Ellis Horwood Series in Mechanical Engineering, Ellis Horwood, 1989.

\bibitem{rangwala91b}
S.~Rangwala and D.~A. Dornfeld,
\newblock {A Study of Acoustic Emission Generated during Orthogonal Metal
  Cutting---2: Spectral Analysis},
\newblock International Journal of Mechanical Sciences {\bf 33}(6), 489--499
  (1991).

\bibitem{holman02}
J.~P. Holman,
\newblock {\em {Heat Transfer}},
\newblock {McGraw}-Hill Series in Mechanical Engineering, McGraw-Hill, 9th
  edition, 2002.

\bibitem{hull84}
D.~Hull and D.~J. Bacon,
\newblock {\em {Introduction to Dislocations}}, volume~37 of {\em International
  Series on Materials Science and Technology},
\newblock Pergamon Press, 3rd edition, 1984.

\bibitem{honeycombe84}
R.~W.~K. Honeycombe,
\newblock {\em {The Plastic Deformation of Metals}},
\newblock Edward Arnold, 2nd edition, 1984.

\bibitem{skalskyi03}
V.~R. Skak's'kyi, O.~E. Andreikiv, and O.~M. Serhienko,
\newblock {Investigation of the Plastic Deformation of materials by the
  Acoustic Emission Method (Review)},
\newblock Materials Science {\bf 39}(1), 86--107 (January 2003).

\bibitem{ono05}
K.~Onu,
\newblock {Current Understanding of Mechanisms of Acoustic Emission},
\newblock The Journal of Strain Analysis for Engineering Design {\bf 40}(1),
  1--15 (January 2005).

\bibitem{kumar68}
A.~Kumar, F.~E. Hauser, and J.~E. Dorn,
\newblock {Viscous Drag on Dislocations in Aluminium at High Strain rates},
\newblock Acta Metallurgica {\bf 16}, 1189--1197 (1968).

\bibitem{kumar69}
A.~Kumar,
\newblock {Strain Rate Effects in Materials},
\newblock Applied Polymer Symposia {\bf 12}, 67--96 (1969).

\bibitem{asm54}
{ASM Committee on Machining},
\newblock {Machining of Steel and Cast Iron},
\newblock Metal Progress, 141--151 (July 1954).

\bibitem{kececioglu58}
D.~Kececioglu,
\newblock {Shear-Strain Rate in Metal Cutting and its Effects on Shear-Flow
  Stress},
\newblock Transactions of the ASME {\bf 80}, 158--168 (January 1958).

\bibitem{ernst41}
H.~Ernst and M.~E. Merchant,
\newblock {Chip Formation, Friction and High Quality Machined Surfaces},
\newblock Transactions of the American Society for Metals {\bf 29}, 299--378
  (1941).

\bibitem{oxley63}
P.~L.~B. Oxley and M.~J.~M. Welsh,
\newblock {Calculating the Shear Angle in Orthogonal Metal Cutting from
  Fundamental Stress, Strain, Strain-Rate Properties of the Work Material},
\newblock in {\em {Proceedings of the 4th International Machine Tool Design and
  Research Conference}}, pages 73--86, 1963.

\bibitem{wright77}
P.~K. Wright and J.~L. Robinson,
\newblock {Material Behaviour in Deformation Zones of Machining Operation},
\newblock Metals Technology {\bf 4}(5), 240--248 (May 1977).

\bibitem{stevenson70}
M.~G. Stevenson and P.~L.~B. Oxley,
\newblock {An Experimental Investigation of the Influence of Strain-Rate and
  Temperature on the Flow Stress Properties of a Low Carbon Steel using a
  Machining Test},
\newblock Proceedings of the Institution of Mechanical Engineers {\bf 185},
  741--754 (1970--1971).

\bibitem{hastings74}
W.~F. Hastings, P.~L.~B. Oxley, and M.~G. Stevenson,
\newblock {Predicting a Materials's Machining Characteristics Using Flow Stress
  Properties Obtained from High-Speed Compression Tests},
\newblock Proceedings of the Institution of Mechanical Engineers {\bf 188},
  245--252 (1974).

\bibitem{zorev63}
N.~N. Zorev,
\newblock {Interrelationship between Shear Process Ocurring along Tool Face and
  on Shear Plane in Metal Cutting},
\newblock in {\em International Research in Production Engineering}, pages
  42--49, 1963.

\bibitem{trent84}
E.~M. Trent,
\newblock {\em {Metal Cutting}},
\newblock Butterworths, 2nd edition, 1984.

\bibitem{kobayashi60}
S.~Kobayashi and E.~G. Thomsen,
\newblock {The role of Friction in Metal Cutting},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  82}, 324--332 (November 1960).

\bibitem{fenton68}
R.~G. Fenton and P.~L.~B. Oxley,
\newblock {Mechanics of Orthogonal Machining: Allowing for the Effects of
  Strain Rate and Temperature on Tool-Chip Friction},
\newblock Procedings of the Institution of Mechanical Engineers {\bf 183},
  417--438 (1968-1969).

\bibitem{xiaoping97}
L.~I. Xiaoping,
\newblock {Development of a Predictive model for Stress Distributions at the
  Tool-Chip Interface in Machining},
\newblock Journal of Materials Processing Technology {\bf 63}, 169--174 (1997).

\bibitem{buryta98}
D.~E. Buryta and R.~Sowerby,
\newblock {Experimental Determination of Rake face Stress Distributions},
\newblock Key Engineering Materials {\bf 138--140}, 57--125 (1998).

\bibitem{hosford83}
W.~F. Hosford and R.~M. Caddell,
\newblock {\em {Metal Forming: Mechanics and Metallurgy}},
\newblock Prentice-Hall, 1983.

\bibitem{bickel63}
E.~Bickel,
\newblock {The Temperature on a Turning Tool},
\newblock in {\em International Research in Production Engineering}, pages
  89--94, 1963.

\bibitem{silva99}
M.~B. Silva and J.~Wallbank,
\newblock {Cutting Temperature: Prediction and Measurement Methods---A Review},
\newblock Journal of Materials Processing Technology {\bf 88}, 195--202 (1999).

\bibitem{stephenson93}
D.~A. Stephenson,
\newblock {Tool-Work Thermocouple Temperature Measurements---Theory and
  Implementation Issues},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  115}(4), 432--437 (November 1993).

\bibitem{chow88}
J.~G. Chow and P.~K. Wright,
\newblock {On-Line Estimation of Tool/Chip Interface Temperatures for a Turning
  Operation},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  110}(1), 56--64 (February 1988).

\bibitem{chao61}
B.~T. Chao, H.~L. Li, and K.~L. Trigger,
\newblock {An Experimental Investigation of Temperature Distribution at
  Tool-Flank Surface},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  83}, 496--504 (November 1961).

\bibitem{jaspers98}
S.~P. F.~C. Jaspers, J.~H. Dautzemberg, and D.~A. Taminiau,
\newblock {Temperature Measurement in Orthogonal Metal Cutting},
\newblock International Journal of Advanced Manufacturing Technology {\bf 14},
  7--12 (1998).

\bibitem{stephenson91}
D.~A. Stephenson,
\newblock {Assessment of Steady-State Metal Cutting Temperature Models Based on
  Simultaneous Infrared and Thermocouple Data},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  113}(2), 121--128 (May 1991).

\bibitem{lee51}
E.~H. Lee and B.~W. Shaffer,
\newblock {The Theory of Plasticity Applied to a Problem of Machining},
\newblock Journal of Applied Mechanics {\bf 18}, 405--413 (December 1951).

\bibitem{kobayashi59}
S.~Kobayashi and E.~G. Thomsen,
\newblock {Some Observations on the Shearing Process in Metal Cutting},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  81}, 251--262 (August 1959).

\bibitem{oxley98}
P.~L.~B. Oxley,
\newblock {Development and Application of a Predictive Machining Theory},
\newblock Machining Science and Technology {\bf 2}, 165--189 (1998).

\bibitem{hastings80}
W.~F. Hastings, P.~Mathew, and P.~L.~B. Oxley,
\newblock {A Machining Theory for Predicting Chip Geometry, Cutting Forces etc.
  from Work Material Properties and Cutting Conditions},
\newblock Proceedings of the Royal Society of London {\bf A371}(1747), 569--587
  (August 1980).

\bibitem{trent59}
E.~M. Trent,
\newblock {Tool Wear and Machinability},
\newblock Institution of Production Engineers Journal {\bf 38}(105), 105--130
  (1959).

\bibitem{spanner74}
J.~C. Spanner,
\newblock {\em {Acoustic Emission: Techniques and Applications}},
\newblock Intex Publishing Company, 1974.

\bibitem{ivanov84}
V.~Ivanov,
\newblock {Acoustic Emission: Some Problems, Tasks and Solutions},
\newblock NDT International {\bf 17}(6), 323--328 (December 1984).

\bibitem{gillis72}
P.~P. Gillis,
\newblock {Dislocation Motions and Acoustic Emissions},
\newblock Acoustic Emission, ASTM STP 505, 20--29 (1972).

\bibitem{rouby77}
D.~Rouby, P.~Fleischmann, and P.~Gobin,
\newblock {An Acoustic Emission Source Model Based on Dislocation Movement},
\newblock in {\em Proceedings of the 6th International Conference on Internal
  Friction and Ultrasonic Attenuation in Solids}, pages 811--815, 1977.

\bibitem{rangwala91a}
S.~Rangwala and D.~A. Dornfeld,
\newblock {A Study of Acoustic Emission Generated during Orthogonal Metal
  Cutting---1: Energy Analysis},
\newblock International Journal of Mechanical Sciences {\bf 33}(6), 471--487
  (1991).

\bibitem{blitz71}
J.~Blitz,
\newblock {\em {Ultrasonics: Methods and Applications}},
\newblock Butterworths, 1971.

\bibitem{diniz92}
A.~E. Diniz, J.~J. Liu, and D.~A. Dornfeld,
\newblock {Correlation Tool Life, Tool Wear and Surface Roughness by Monitoring
  Acoustic Emission in Finish Turning},
\newblock Wear {\bf 152}, 395--407 (1992).

\bibitem{kannatey-asibu81}
E.~Kannatey-Asibu, Jr. and D.~A. Dornfeld,
\newblock {Quantitative Relationships for Acoustic Emission from Orthogonal
  Metal Cutting},
\newblock Journal of Engineering for Industry---Transactions of the ASME {\bf
  103}(3), 330--340 (August 1981).

\bibitem{moriwaki83}
T.~Moriwaki,
\newblock {Application of Acoustic Emission Measurement to Sensing of Wear and
  Breakage of Cutting Tool},
\newblock Bulletin of the Japan Society of Precision Engineering {\bf 17},
  154--160 (September 1983).

\bibitem{dornfeld80}
D.~A. Dornfeld and E.~Kannatey-Asibu, Jr.,
\newblock {Acoustic Emission during Orthogonal Metal Cutting},
\newblock International Journal of Mechanical Science {\bf 22}(5), 285--296
  (1980).

\bibitem{schmenk84}
M.~J. Schmenk,
\newblock {Acoustic Emission and the Mechanics of Metalcutting},
\newblock in {\em Acoustic Emission Monitoring and Analysis in
  Manufacturing---Winter Annual Meeting of the {ASME}}, pages 95--106, December
  1984.

\bibitem{lan86}
M.~S. Lan and D.~A. Dornfeld,
\newblock {Acoustic Emission and Machining---Process Analysis and Control},
\newblock Advanced Manufacturing Processes {\bf 1}(1), 1--21 (1986).

\bibitem{saini96}
D.~P. Saini and Y.~J. Park,
\newblock {A Quantitative Model of Acoustic Emission in Orthogonal Cutting
  Operations},
\newblock Journal of Materials Processing Technology {\bf 58}, 343--350 (1996).

\bibitem{lan85}
M.~S. Lan and Y.~Naerheim,
\newblock {Application of Acoustic Emission Monitoring in Machining},
\newblock in {\em Proceedings of the 13th North American Manufacturing Research
  Conference}, pages 310--313, May 1985.

\bibitem{grabec77}
I.~Grabec and P.~Leskovar,
\newblock {Acoustic Emission of a Cutting Process},
\newblock Ultrasonics {\bf 15}(1), 17--20 (January 1977).

\bibitem{yaohui88}
Z.~Yaohui and X.~Rongbao,
\newblock {A Study on the Separation of Acoustic Emission Signals in Metal
  Cutting Process},
\newblock in {\em Progress in Acoustic Emission {IV}---The Japanese Society for
  {NDI}}, pages 462--469, 1988.

\bibitem{lin82}
G.~C.~I. Lin, P.~Mathew, P.~L.~B. Oxley, and A.~R. Watson,
\newblock {Predicting Cutting Forces for Oblique Machining Conditions},
\newblock Proceedings of the Institute of Mechanical Engineers {\bf 196},
  141--148 (1982).

\bibitem{adibi-sedeh03}
A.~H. Adibi-Sedeh, V.~Madhavan, and B.~Bahr,
\newblock {Extension of Oxley's Analysis of Machining to Use Different Material
  Models},
\newblock Journal of Manufacturing Science and Engineering---Transactions of
  the ASME {\bf 125}(4), 656--666 (November 2003).

\bibitem{oyane67}
M.~Oyane, F.~Takashima, K.~Osakada, and H.~Tanaka,
\newblock {The Behaviour some Steels under Dynamic Compression},
\newblock 10th Japan Congress on Testing Materials, 72--76 (1967).

\bibitem{macgregor46}
C.~W. MacGregor and J.~C. Fisher,
\newblock {A Velocity-Modified Temperature for the Plastic Flow of Metals},
\newblock Journal of Applied Mechanics {\bf 13}, A11--A16 (March 1946).

\end{thebibliography}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
Acoustic emission (AE) refers to the elastic stress waves
generated at a source by some sudden release of strain energy
within a material due to a rearrangement of its internal structure
\cite{spanner74,kannatey-asibu81}. In metal cutting, there are two
major processes that produce changes in the internal structure of
the work material: dislocation motion, which results in plastic
deformation, and fracture. It has been
proposed~\cite{kannatey-asibu81,moriwaki83} that the sources that
contribute to AE generation during metal cutting are the following
(Figure~\ref{fig:ae_zones}):
\begin{itemize}
   \item Plastic deformation in the primary and secondary zones.
   \item Tool-chip and tool-work sliding friction in the secondary
   and tertiary zones respectively.
   \item Colision, entangling and fracture of chips.
   \item Fracture of the cutting tool.
\end{itemize}
\begin{figure}[h]
   \center{\includegraphics{ae_zones.eps}}
   \caption[Sources of AE in metal cutting.]
   {Sources of AE in metal cutting. (a)~Continuous chip formation.
   (b)~Discontinuous chip formation.}
   \label{fig:ae_zones}
\end{figure}
Moreover, the AE emissions have been reported to be dependent on
work and tool materials, tool geometry, cutting speed, feed rate,
width of cut and cutting fluids.

The stress waves generated at the emission source propagate
through the material at roughly the speed of sound as shear and
pressure waves. Surface wave AE propagation is also possible. As
the stress waves propagate, they are attenuated, reflected and
distorted. At the surface of the material the amplitude of the AE
vibrations is very small (usually a few nanometres) and the
frequency content may extend from the audible frequency zone to
the megahertz range. However, in metal cutting, the frequency
range of the signals normally utilized is in the order of
\mbox{$10^2$--$10^3$ kHz}.

Acoustic emission can be detected with an appropriate transducer,
coupled to the surface of the material, that, when stimulated by
the AE stress waves, transforms the mechanical excitations into
electrical signals. Usually, AE transducers are constructed of a
piezoelectric ceramic element, lead zirconate titanate (PZT).
Figure~\ref{fig:pzt_transducer} shows a conventional PZT
transducer.
\begin{figure}[h]
   \center{\includegraphics{pzt_transducer.eps}}
   \caption{Piezoelectric transducer.}
   \label{fig:pzt_transducer}
\end{figure}
The detected signals are then amplified, filtered, and transmitted
to equipment that stores and analyses the incoming signals. In
metal cutting, common amplification factors range from
$20$--$60$~dB, and band-pass filters are usually employed in the
$10^2$--$10^3$~kHz range. Figure~\ref{fig:ae_process_diagram}
shows the various stages of the AE process, starting from the
emission source to the final stages of signal processing and
analysis.


Due to the randomness of the AE signals, the analysis and
interpretation of AE is normally quite complex. Therefore, signal
processing techniques are normally needed to process the raw AE
signals. The principal processing techniques applied to AE
generated in metal cutting are described next:

\begin{description}

   \item [Energy representation] The most common way to represent
   the energy of a signal is by calculating its root mean square
   (RMS). This is done by taking the square root of the mean of
   the square of the signal time series. Another method makes use
   of the previously filtered and full-wave rectified signal that
   is then used to compute a probability density function and to
   compute the maximum value of the distribution function, the mode.
   The mode is known to be a statistical averaging technique that
   is less sensitive to outlying values than the mean.

   \item [Count techniques] There are two major techniques: total
   counts ($N$) and count rate ($\dot N$). The total counts is the
   number of times a signal overcomes a given threshold. The
   number of counts per unit time is the count rate. The
   dependence of the count techniques on a threshold level can lead
   to problems since if the signal amplitude changes, the count
   becomes meaningless if the threshold in also not changed.

   \item [Spectra analysis] When a signal is sampled, it is
   normally represented as a series of amplitudes. The signal is
   said to be represented in the time domain, i.e.\ the signal is
   represented as a function of time. The signal can also be
   represented as a function of frequency (frequency domain
   representation). Normally, the fast Fourier
   transform (FFT) algorithm is applied to compute the transformation
   from the time to the frequency domain. The plot showing the
   distribution of the power of the signal in the frequency domain
   is known as the power spectrum.

\end{description}

Acoustic emission is normally classified into two types of
signals: continuous and burst (transient) type. Continuous signals
arrive at the transducer in large numbers and thus distinct events
can not be detected, whereas burst type signals are characterized
by distinct events (Figure~\ref{fig:ae_types}).
\begin{figure}[h]
   \center{\includegraphics{ae_types.eps}}
   \caption{Continuous and burst type acoustic emission.}
   \label{fig:ae_types}
\end{figure}
In turning, continuous type AE is normally attributed to
deformation in the primary and secondary zones, and friction in
the secondary and tertiary zones~\cite{lan86}, while burst type AE
is attributed to chip and tool fracture \cite{dornfeld83}.
Additionally, chip impact and tangling also generate burst type AE
signals \cite{inasaki98}.

The major advantage of AE over other signals is that its frequency
range is much higher than that of the machine vibrations and other
environmental noises. Moreover, it is expected that AE is very
sensitive to metal cutting process characteristics since it is
generated by micro-mechanical activities in the cutting zone.


[Propagation of AE though the primary shear plane and tool-chip
interface \cite{uehara84}.] [pencil lead breaking test]
